1/1

Jour Fixe

WFP, PG, SS, NJ, kHz, NW, TK

General

- Both labs will prepare to make the labs holiday ready.
- Philipp will exchange old Benzene for new.

Atomslab

- Most orders from Mg wishlist are done. Some more remain and will be finished by end of this week.
- Ausschreibung process has been started. Herr Barkow needs OK on a few documents and then everything goes online. Tender will be open till 7th January and then we can order the laser 4 weeks after that.
- Quench system
 - $\,\circ\,$ Worked a bit on AOM in coupling. Beam is perhaps too big for AOM but will leave it as it is at the moment.
 - $\circ\,$ SHG lock had some offset problems. Seems to be solved for the moment.

ResLab

• PCF in coupling has been not possible yet. Fiber pen light is coupled in cladding from behind, but don't yet see any light through the core. Could possibly ask Menlo to install it (Available dates 17th January).

